\documentclass[12pt]{amsart}

\setlength{\parindent}{0pt}
\setlength{\textheight}{8.5in}
\setlength{\voffset}{-1in}
\setlength{\textwidth}{6.5in}
\setlength{\hoffset}{-.5in}
\everymath={\displaystyle}
\newcommand{\C}{\ensuremath{\mathbb{C}}}
\pagestyle{empty}

\begin{document}
\thispagestyle{empty}

\textbf{Math 425: Complex Variables}
\hfill\makebox[2.5in][r]{Name:\hrulefill}

\textbf{Quiz 1}\quad \textsl{Jan. 25, 2002}

\bigskip
1. Find the roots of $f(z) = z^4 + 1$.

\vfill

2. Use DeMoivre's Formula to compute $\big(1 + i\sqrt{3}\,\big)^9$.

\vfill

3. Define the boundary $\partial D$ of a set $D\subset\C$.

\vfill

\end{document}
